Author's response to reviews

Title: Treatment of Periprosthetic Femoral Fractures after Femoral Revision Using a Long Stem

Authors:

Youngwoo Kim (woochan76@hotmail.co.jp)
Chiaki Tanaka (c.tanaka@kyoto.zaq.jp)
Hiroshi Tada (tadah4416@gmail.com)
Hiroshi Kanoe (kanoehiroshi@gmail.com)
Takaaki Shirai (takaakis@kuhp.kyoto-u.ac.jp)

Version: 7 Date: 22 April 2015

Author's response to reviews: see over
Dear Dr. Catia

1) Please clarify whether the consent was written or verbal.

We have received the informed consent verbally. We have informed consent to patients for use of their data in this study.

Sincerely,

Youngwoo Kim, M.D., Ph.D. (Corresponding author)
Department of Orthopaedic Surgery
Kyoto City Hospital
1-2, Higashitakada-cho, Mibu, Nakagyo-ku
Kyoto, 604-8845, Japan
Tel: 81-75-311-5311
Fax: 81-75-321-6025
E-mail: woochan76@hotmail.co.jp

Chiaki Tanaka, MD, PhD
Department of Orthopaedic Surgery
Kyoto City Hospital
1-2, Higashitakada-cho, Mibu, Nakagyo-ku
Kyoto, 604-8845, Japan
Tel: +81-75-311-5311
Fax: +81-75-321-6025

Hiroshi Tada, MD,
Department of Orthopaedic Surgery
Kyoto City Hospital
1-2, Higashitakada-cho, Mibu, Nakagyo-ku
Kyoto, 604-8845, Japan
Hiroshi Kanoe, MD, PhD
Department of Orthopaedic Surgery
Kyoto City Hospital
1-2, Higashitakada-cho, Mibu, Nakagyo-ku
Kyoto, 604-8845, Japan
Tel: +81-75-311-5311
Fax: +81-75-321-6025

Takaaki Shirai, MD, PhD
Department of Orthopaedic Surgery
Kyoto City Hospital
1-2, Higashitakada-cho, Mibu, Nakagyo-ku
Kyoto, 604-8845, Japan
Tel: +81-75-311-5311
Fax: +81-75-321-6025